Author's response to reviews

Title: Intravitreal vs. Subtenon Triamcinolone Acetonide for the treatment of diabetic cystoid macular edema

Authors:

Mauro Cellini (mauro.cellini@unibo.it)
Alberto Pazzaglia (angela.ungarelli@unibo.it)
Eugenio Zamparini (ezampar@tin.it)
Pietro Leonetti (pietro.leonetti@yahoo.it)
Emilio C Campos (emilio.campos@unibo.it)

Version: 5 Date: 20 February 2008

Author's response to reviews: see over
Bologna, February 21st, 2008

Robin Cassady-Cain
Assistant Editor
BMC-series journals

Object: article MS: 1977705101167421

Dear Dr. Cassidy-Cain,

here enclosed you will find a point-by-point replay to the reviewers’ comments and the revision version of the manuscript entitled: **Intravitreal vs. Subtenon Triamcinolone Acetonide for the treatment of diabetic cystoid macular edema** (ISRCTN67086909).

Makoto Inoue

1. The English was reviewed by an english mother tongue of the John Hopkins University
2. The authors proposed a semplified technique for the posterior subtenon injection of triamcinolone acetonide and they show by B-scan echography the correct placement, with this technique, of triamcinolone acetonide near the sclera in the macular region.

I hope that the revision version it is O.K. for a positive judgement of the referees.

Sincerely yours

Dr. Mauro Cellini